Title: Virologic suppression in response to antiretroviral therapy despite extensive resistance within HIV-1 reverse transcriptase after the first virologic failure

Authors:

Marta Iglis de Oliveira (mariglis@hotmail.com)

VALTER SOUZA JUNIOR (jr_walter@hotmail.co.uk)

Claudia Fernanda de Lacerda Vidal (vidal.claudia@gmail.com)

Paulo Sérgio Ramos de Araújo (psergiora@gmail.com)

Version: 1 Date: 24 Mar 2018

Author’s response to reviews:

to Editor's comments:

1. informed consent was not obtained from all participants because it was waived by the ethics committee of Federal University of Pernambuco. Furthermore, the data was also de-identified.

to Editorial Policies

1-For the 'Availability of data and materials' section, WE provided information about where the data supporting ours findings can be found.